Author's response to reviews

Title: Design of an online health-promoting community: negotiating user community needs with public health goals and service capabilities

Authors:

Joakim Ekberg (joakim.ekberg@liu.se)
Toomas Timpka (toomas.timpka@liu.se)
Marianne Angbratt (marianne.angbratt@lio.se)
Linda Frank (linda.frank@lj.se)
Anna-Maria Norén (annamaria.noren@ltkalmar.se)
Lena Hedin (lena.hedin@lj.se)
Emelie Andersen (emelie.andersen@lj.se)
Elin A Gursky (elin.gursky@anser.org)
Boel Andersson Gäre (boel.andersson.gare@lj.se)

Version: 2 Date: 22 October 2012

Author's response to reviews:

Dear Editor,

We are grateful for the opportunity to correct the manuscript. In accordance with your recommendation, the following was added to the method section, at the very end of the section data collection:

"The project was approved by the Ethics Committee of Linköping University, Dnr: 479-31."

The author contribution formatting was corrected to read:

"JE participated in the design of the study, performed the analysis and drafted the manuscript. TT participated in the design of the study, performed the analysis and drafted the manuscript. MA participated in the design of the study, performed the analysis and revised the manuscript. LF participated in the design of the study, performed the analysis and revised the manuscript. AN participated in the design of the study, performed the analysis and revised the manuscript. LH participated in the design of the study, performed the analysis and revised the manuscript. EA participated in the design of the study, performed the analysis and revised the manuscript. EAG participated in the design of the study, performed the analysis and revised the manuscript. BA participated in the design of the study, performed the analysis and revised the manuscript. All authors read and approved the final manuscript."

Best regards

Joakim